Author's response to reviews

Title: Obesity and elevated blood pressure among adolescents in Lagos, Nigeria: A cross-sectional study.

Authors:

Abiola O Oduwole (bioodu@yahoo.com)
Taiwo A Ladapo (drteeladapo@yahoo.com)
Iretiola B Fajolu (iretifajolu@justice.com)
Ekanem N Ekure (ekaekure@yahoo.com)
Olufunmilayo Funke Adeniyi (layo_funke@yahoo.co.uk)

Version: 3 Date: 29 June 2012

Author's response to reviews:

Research article: Obesity and elevated blood pressure among adolescents in Lagos, Nigeria: A cross-sectional study.

Response to Reviewers Comments:
1. Spaces between words and numbers have been appropriately adjusted.
   -Results: page 2, line 3
   -Page 4, line 7
   -others as deemed appropriate

2. Uniformity has been adhered to for expression of significance.
   = when probability is exact and < or > in other situations as appropriate.

3. Pearson correlation coefficient between BMI and weight in females has been corrected.

4. The relevant corrections in the references have been made.

5. Manuscript has been edited by Edanz editing group.